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Communications problems continue to plague the R/V Brooks McCall.  Internet and e-mail connections are very sporadic, but functional.  Handheld Iridium phones are the most dependable method of voice communication.  All the different scientific groups on board are working very well together and the ship crew has been extremely supportive.
We continue to implement the modified sampling regime implemented on 05/30/10, in order to accommodate microbial analyses of samples from each cast.  Chemistry samples will be taken from odd CTD bottle numbers (1,3,5,7,9) and biology from even numbers (2,4,6,8,10), including Rototox.  Bottle 99 is a surface bucket sample.  
Based on results of sampling conducted on 5/30/10, we determined that the best way to fill existing information gaps would be to attempt to follow the plume away from the well head by sampling a WSW transect that intersected the sample location with the highest fluorescence (BM 53) and sampling at 5km intervals.

The first cast at the 5km WSW transect station failed to detect significant fluorescence.  A decision was made to transit south, on the same 5km arc, to attempt to intersect the plume along a SW transect from the well head.  The second cast (BM57) found a high fluorescence signature at depths of 1100 to 1200m, and inversely correlated changes in Dissolved Oxygen.  A smaller fluorescence signature was also noted near the bottom at 1500m.  This appears to indicate that as oil migrates farther from the source, microbial activity is degrading the dispersed plume and consuming the oxygen.  Samples for microbial analysis were collected.
Based on the findings at BM57, we travelled outward on the same SW transect to 10km from the source.  At that distance, BM58 showed two narrow, very distinct peaks at depths of 1130 and 1180m, with strong corresponding dissolved oxygen reductions at both peaks.  A smaller peak was detected at 1280m that was not associated with a drop in DO.  Samples were collected for the entire suite of analyses from all three depths.  
We continued to follow the same SW transect out to 15km, for station BM59, and did not find any significant fluorescence at any depth.

On 6/01/10, we will follow a SSW heading, starting at 1.5km from the well head.
Rototox testing is continuing on selected samples from each cast, but has been delayed by hatching failure of the first batch of organisms on this leg.  Subsequent hatches have been successful. Results for BM 53 and BM 54 will be provided in the 6/01/10 report, and BM 55 thru BM 59 should be provided in the 6/02/10 report.
